**Author’s response to reviews**

**Title:** The role of T2* Gradient Echo in the diagnosis of tumefactive intrahepatic extramedullary hematopoiesis in myelodysplastic syndrome and diffuse hepatic iron-overload: A case report and literature review.

**Authors:**

abel belay (abelassefamd@gmail.com)

Andrew Bellizzi (andrew-bellizzi@uiowa.edu)

Alan Stolpen (alan-stolpen@uiowa.edu)

**Version: 2 Date:** 24 Oct 2017

**Author’s response to reviews:**

Response to reviewer 1: Blood smear added.

Response to reviewer 2: 15 pack-years of smoking history and no significant family history were added to the manuscript.

Comparing to another case report would be difficult specially since my case report is on the T2* GRE characteristics of EMH in the setting of secondary hemochromatosis, and as far as I know I am describing the first case (no literature on this topic).